Author's response to reviews

Title: A pilot study of Aboriginal health promotion from an ecological perspective

Authors:

Rachel E Reilly (rachel@unimelb.edu.au)
Marion Cincotta (marionc@vaccho.com.au)
Joyce Doyle (jdoyle6@bigpond.com)
Bradley R Firebrace (bradleyf@unimelb.edu.au)
Margaret Cargo (margaret.cargo@unisa.edu.au)
Gemma van den Tol (gvandentol@gvct.com.au)
Denise Morgan-Bulled (denise@vineymorganams.com.au)
Kevin G Rowley (rowleyk@unimelb.edu.au)

Version: 7 Date: 22 August 2011

Author's response to reviews:

Dear Production Team,

We have made the following minor alterations:

p.7: last paragraph, line 3: game-days to game days.
p.17, last paragraph, line 6: Space added before Folate
p.19, last line: KG to KR
REFERENCES:
3. 'Ausstats Catalogue number...' added
21. 'In Press' replaced with citation.
26. Category to Catalogue
27. Group WCH to Woongi Community Health Group
31. 'In.' removed
32. 'In.' removed
Page break added between references and Figure Legend

Legends to Legend

Many thanks,
Rachel Reilly